Author's response to reviews

Title: A Comparison of a Social Support Physical Activity Intervention in Weight Management among Post-partum Latinas

Authors:

Colleen Keller (Colleen.Keller@asu.edu)
Barbara Ainsworth (Barbara.Ainsworth@asu.edu)
Kathryn Records (Kathie.Records@asu.edu)
Michael Todd (Mike.Todd@asu.edu)
Michael Belyea (Michael.Belyea@asu.edu)
Sonia Vega-López (Sonia.Vega.Lopez@asu.edu)
Paska Permana (Paska.Permana@va.gov)
Dean Coonrod (Dean_Coonrod@dmgaz.org)
Allison Nagle-Williams (Allison.Nagle@asu.edu)

Version: 4 Date: 28 August 2014

Author's response to reviews: see over
August 28, 2014

Dr. Proel Hennessy Vargas
Journal Editorial Office
BioMed Central
Floor 6, 236 Gray’s Inn Road
London
WC1X 8HB
United Kingdom
e-mail: editorial@biomedcentral.com

Dear Dr. Proel Hennessy Vargas,

Thank you for our long awaited and excellent review for A Comparison of a Social Support Physical Activity Intervention in Weight Management among Post-partum Latinas submitted by Colleen Keller, Barbara Ainsworth, Kathryn Records, Michael Todd, Michael Belyea, Sonia Vega-López, Paska Permana, Dean Coonrod, and Allison Nagle-Williams; BMC Public Health.

Below are the reviewer comments and our corrections; all have strengthened and clarified the report. Thank you for the helpful and supportive comments. We look forward to your editorial response.

<table>
<thead>
<tr>
<th>Reviewer 1</th>
<th>Thank you for this correction. We have corrected the text.</th>
</tr>
</thead>
<tbody>
<tr>
<td><strong>Major Compulsory Revisions:</strong></td>
<td>In the report of Results pertaining to Energy Intake,</td>
</tr>
<tr>
<td></td>
<td>the authors state the control group had a higher</td>
</tr>
<tr>
<td></td>
<td>saturated fat intake than the intervention group, but</td>
</tr>
<tr>
<td></td>
<td>the data in Table 3 indicate that these results are</td>
</tr>
<tr>
<td></td>
<td>actually the opposite: the control group %saturated fat</td>
</tr>
<tr>
<td></td>
<td>was 10.1, 9.9, and 9.5% versus the intervention group</td>
</tr>
<tr>
<td></td>
<td>saturated fat% of 10.6, 10.3, and 11.0%. Thus, the text</td>
</tr>
<tr>
<td></td>
<td>should be corrected.</td>
</tr>
<tr>
<td><strong>Discretionary Revisions:</strong></td>
<td>The paragraphs are now joined.</td>
</tr>
<tr>
<td></td>
<td>The authors stated they joined the first 2 paragraphs in</td>
</tr>
<tr>
<td></td>
<td>the Background, but seemed to have omitted doing so.</td>
</tr>
</tbody>
</table>

Sincerely,

Colleen Keller, PhD, FAHA, FAAN  
*Regents’ Professor*
*Foundation Professor in Women’s Health and*
*Director, Center for Healthy Outcomes in Aging*
*Co-Director, T32 Transdisciplinary Training in Health Disparities Science*